Author's response to reviews

Title: Identifying Quality Improvement intervention publications - A comparison of electronic search strategies

Authors:

Susanne Hempel (susanne_hempel@rand.org)
Lisa Rubenstein (lisa_rubenstein@rand.org)
Roberta Shanman (Roberta_shanman@rand.org)
Robbie Foy (R.Foy@leeds.ac.uk)
Su Golder (su.golder@york.ac.uk)
Danz Marjorie (mjsdanz@gmail.com)
Shekelle Paul (paul_shekelle@rand.org)

Version: 4 Date: 12 July 2011

Author's response to reviews:

Hello,

I've just uploaded the copy-edited version of the file.

I did not substitute 'quality improvement' for 'QI' throughout the text as sometimes the abbreviation will just not do, I hope that is ok.

I also did not accept the change in the tables which substituted numerical numbers for the written word, this did not seem to make sense and I am referencing the strategies throughout the text where they are still called strategy #2 etc. I could not change the search terms in the tables (e.g. you prefer 'healthcare' to 'health care') because these were the actual terms used and I had to leave the spaces in the search strategies to distinguish the search strategy number and the numerical replacement of individual search terms.

There was one format change for table 2 that I did not understand, when I accepted it took the border away of seemingly random cells of the table (some but not all). Please let me know if tables should not have borders at all, then I'll change the tables.

Many thanks for your help,

Susanne.